Author's response to reviews

Title: Frequency Format Diagram and Probability Chart for Breast Cancer Risk Communication: a Prospective, Randomized Trial

Authors:

Karthik Ghosh (ghosh.karthik@mayo.edu)
Brianna J Crawford (crawford.brianna@mayo.edu)
Sandhya Pruthi (pruthi.sandhya@mayo.edu)
Constance I Williams (williams.constance@mayo.edu)
Lonzetta Neal (neal.lonzetta@mayo.edu)
Nicole P Sandhu (sandhu.nicole@mayo.edu)
Ruth E Johnson (johnsonmd.ruth@mayo.edu)
Dietlind Wahner-Roedler (wahnerroedler.dietlind@mayo.edu)
Marcia K Britain (britain.marcia@mayo.edu)
Stephen S Cha (cha@mayo.edu)
Amit K Ghosh (ghosh.amit@mayo.edu)

Version: 5 Date: 13 October 2008

Author's response to reviews: see over
October 13, 2008

The Biomed Central Editorial Production Team
34-42 Cleveland Street
London W1T 4LB
United Kingdom
London W1T 4LB

Dear Editorial Production Team:

The enclosed revised manuscript #1656132309189872


has been reformatted per your instruction for BioMed Central Medical Education.

This manuscript has been resubmitted to your journal electronically.

Mayo Foundation for Medical Education and Research is the author for the purpose of copyright ownership under the work-for-hire provision of the copyright law. The authorized agent transfers that portion of the copyright that applies to the Mayo Foundation authors. Your transfer agreement will be signed upon receipt.

Address reprint requests to Karthik Ghosh, MD, Breast Diagnostic Clinic, Mayo Clinic, 200 First Street SW, Rochester, MN 55905, USA. E-mail: ghosh.karthik@mayo.edu.

Sincerely,

Karthik Ghosh, MD, MS

KG:
Enc.: manuscript; 3 figures; authors’ response to formatting requests